Author's response to reviews

Title: 3D-Ultrasonography for Evaluation of Facial Muscles in Patients with Chronic Facial Palsy or Defective Healing: a Pilot Study

Authors:

Gerd F Volk (fabian.volk@med.uni-jena.de)
Martin Pohlmann (martin.pohlmann87@web.de)
Mira Finkensieper (mira.finkensieper@med.uni-jena.de)
Heather J Chalmers (heather.chalmers@uoguelph.ca)
Orlando Guntinas-Lichius (orlando.guntinas@med.uni-jena.de)

Version: 2 Date: 16 September 2013

Author's response to reviews:

Thank you for your email. You had 2 requests:

1. This is NOT a clinical trial needing a trial registration.
   We report on a pilot study which may lead to a clinical trial!
   Also confirmed by the ethics committee consent this is not needed for such projects!

2. Consent: text was changed.

3. Full name and affiliation of the institutional ethics committee: added.